Author's response to reviews

Title: A Strategy to enhance the Safety and Efficiency of Handovers of ICU Patients: Study Protocol of the pICUp Study

Authors:

Nelleke van Sluisveld (n.vansluisveld@iq.umcn.nl)
Marieke Zegers (m.zegers@iq.umcn.nl)
Gert Westert (g.westert@iq.umcn.nl)
Johannes G van der Hoeven (j.vanderhoeven@ic.umcn.nl)
Hub Wollersheim (h.wollersheim@iq.umcn.nl)

Version: 4 Date: 10 June 2013

Author's response to reviews: see over
April 10, 2013

Dear editor,

Thank you very much for editing our manuscript. Enclosed you will find our manuscript adapted to almost all your proposed changes.

We rejected the deletion of the enumeration under the heading ‘theoretical framework’ on page 7, because these refer to an enumeration in Figure 1:

“The domains are related to the: 1) characteristics of the intervention; 2) societal context; 3) characteristics of the implementation efforts; 4) characteristics of the healthcare facility; 5) social context (e.g., interpersonal, interdepartmental and interinstitutional relationships); 6) professional characteristics; and 7) patient characteristics.”

We are looking forward to hearing whether you agree with this rejection.

Yours sincerely, on behalf of all co-authors,

Nelleke van Sluisveld, MSc

IQ healthcare, Radboud University Nijmegen Medical Centre
Geert Grootplein 21
PO box 9101
6500 HB Nijmegen
The Netherlands
Tel: +31 (0)24 366 73 07
Fax: +31 (0)24 354 01 66
Email: n.vansluisveld@iq.umcn.nl